

Research Article



Check for updates

Temporal Order of Sexual Offending Is Risk-Relevant for Individuals With Child Sexual Exploitation Materials Offences

Kelly M. Babchishin^{1,2} ^(b), Seung C. Lee³ ^(b), Angela W. Eke⁴, Michael C. Seto² ^(b)

 Department of Psychology, Carleton University, Ottawa, Ontario, Canada.
 University of Ottawa's Institute of Mental Health Research, The Royal Ottawa Health Care Group, Ottawa, Ontario, Canada.
 Public Safety Canada, Ottawa, Ontario, Canada.
 Criminal Behaviour Analysis Section, Ontario Provincial Police, Orillia, Ontario, Canada.

Sexual Offending: Theory, Research, and Prevention, 2022, Vol. 17, Article e7229, https://doi.org/10.5964/sotrap.7229

Received: 2021-07-27 • Accepted: 2022-10-03 • Published (VoR): 2022-12-13

Handling Editor: Daniel Turner, University Medical Center Mainz, Mainz, Germany

Corresponding Author: Kelly M. Babchishin, Department of Psychology, Carleton University, 1125 Colonel By Drive, K1S 5B6, Ottawa, Canada. E-mail: Kelly.Babchishin@carleton.ca

Abstract

The current study examined the extent to which the temporal order of sexual offending may be risk-relevant for men with Child Sexual Exploitation Material (CSEM; also referred to as child pornography) offences. We categorized 85 men who had committed two distinct sexual offences (CSEM or contact sexual offence) into three groups: (1) 47% (n = 40) followed a stable pattern, that is, men with CSEM offences who then committed a new CSEM offence; (2) 41% (n = 35) followed a de-escalation pattern, that is, men with contact sexual offences who then committed a CSEM offence; (3) and 12% (n = 10) followed an escalation pattern, that is, men with CSEM offences who then committed a contact sexual offence. Compared to the other groups, the stable group had more sexual interest in children, the de-escalation group had a younger age at first police involvement and more prior offending, and the escalation group had more substance use problems. We then examined recidivism (any new offence after the second sexual offence) and found that the escalation group had the highest 5-year and 7-year reoffending rates (start of follow-up: opportunity after the second sexual offence) for any crime, any non-sexual violence, any violence (including contact sexual offences), and any contact sexual recidivism. The de-escalation and stable groups had the highest CSEM recidivism rates. The current study suggests that ordering of offending within men adjudicated for CSEM offences is risk-relevant and that those who fit the escalation pattern may be at higher risk to reoffend.



This is an open access article distributed under the terms of the Creative Commons Attribution 4.0 International License, CC BY 4.0, which permits unrestricted use, distribution, and reproduction, provided the original work is properly cited.

Keywords

child pornography, child sexual exploitation material, recidivism, risk factor, escalation

Non-Technical Summary

Background

Individuals with Child Sexual Exploitation Materials (CSEM; also referred to legally as child pornography in Canada and the United States) offences now represent a large proportion of caseloads of sexual offences. Among those with both contact sexual offences and CSEM offences, it is not yet known whether the order of sexual offences (i.e., contact then CSEM or CSEM then contact) is related to risk of reoffending.

Why was this study done?

The order of the sexual offences in mixed CSEM offending – whether a CSEM offence was committed before or after a contact sexual offence – and its association with risk factors and future reoffending has not yet been examined.

What did the researchers do and find?

From a sample of 387 men adjudicated for CSEM offences, we categorized the 85 men who had committed two distinct sexual offences (CSEM or contact sexual offence) into three groups: (1) 47% (n = 40) followed a stable pattern, that is, men with CSEM offences who then committed a new CSEM offence; (2) 41% (n = 35) followed a de-escalation pattern, that is, men with contact sexual offences who then committed a CSEM offence; (3) and 12% (n = 10) followed an escalation pattern, that is, men with CSEM offence. We found that men who fit the escalation pattern (CSEM then contact sexual offence) scored higher on risk factors and had higher recidivism rates.

What do these findings mean?

The management and treatment of individuals with CSEM offending should be sensitive to their sexual offending patterns.

Highlights

- Ordering of offending within men adjudicated for Child Sexual Exploitation Materials (CSEM) offences is risk-relevant.
- Men who fit the escalation pattern (CSEM then contact sexual offence) may be at a relatively higher risk to reoffend.
- A significant proportion (8%) of men who fit the stable pattern (CSEM then CSEM) commit a contact sexual offence after 5 and 7 years.
- The management and treatment of individuals with CSEM offending should be sensitive to their sexual offending patterns.



Individuals with Child Sexual Exploitation Materials (CSEM; also referred to legally as child pornography in Canada and the United States) offences represent a large proportion of caseloads of sexual offences seen by police, corrections, community supervision officers, and treatment providers. In Canada, the rate of CSEM reported to police increased by 288% from 2010 to 2017 (Department of Justice Canada, 2019), echoing trends of year after year increases in CSEM cases in other countries (UK: McManus & Almond, 2014; US: Adams & Flynn, 2017). The COVID-19 pandemic has further accelerated CSEM offending (Europol, 2020; Wright, 2020), as well as the use of online legal pornography (Wright, 2020). Research direction in terms of management and sentencing are thus essential to support evidence-based practice.

Individuals with CSEM offences have heterogeneous offending patterns. Approximately 1 in 8 have a prior contact sexual offence (i.e., mixed offence type, 15.4% in Elliott et al., 2019; 12.2% in Seto et al., 2011), and 1 in 10 men with CSEM offences have at least one prior nonsexual offence (11.8%; Babchishin et al., 2011). Self-reported rates of crimes are higher than official rates among men with CSEM offences, with approximately half of men with CSEM offences reporting contact sexual offences (Seto et al., 2011). Of individuals with both CSEM and contact sexual offences, we do not know yet the extent to which the temporal order of sexual offences is risk-relevant. That is, mixed offending individuals could escalate (CSEM to contact) or de-escalate (contact to CSEM). The first group (escalation) is of particular public interest because many policies and laws – such as lengthy sentences for CSEM offences in the US (Hamilton, 2011) are based on the assumption that CSEM is a gateway offence for contact sexual offending (Steiker, 2013). The gateway hypothesis presupposes that participating in lower degree antisocial or problematic behaviours in the future.

Support for the gateway hypothesis – that less severe problematic behaviour can lead to more severe problematic behaviour – has been found for substance use, including tobacco use to illicit drugs (Prescott, 1976), legal and medically prescribed drugs to illegal drugs, as well as from alcohol to illicit drugs (Kandel & Logan, 1984; Yamaguchi & Kandel, 1984). In addition, despite many youths desisting in aggression, an important proportion escalate in severity. Specifically, some boys begin with minor aggression (e.g., bullying), minor delinquency (e.g., shoplifting), or authority problems (e.g., truancy), move to minor violence (e.g., physical fighting) or property damage (e.g., vandalism), and eventually commit serious violence (e.g., sexual offences) or serious delinquency (e.g., burglary; for review, see Loeber & Hay, 1997).

The offending patterns of men with CSEM offences may be meaningful, as individuals with mixed CSEM offending (i.e., individuals who have committed both CSEM and contact sexual offences) have been found to differ from individuals with CSEM-exclusive offending (i.e., those with CSEM offences but no contact sexual offences) and individuals with exclusively contact sexual offences (i.e., those without CSEM offences)



on motivational, facilitative, and situational factors for sexual offending (Babchishin et al., 2015; Elliott et al., 2019; Henshaw et al., 2018). Motivational factors largely relate to having an atypical sexual interest for the sexual behaviour (e.g., pedophilia and accessing CSEM or having sexual contact with children), facilitative factors can increase the likelihood of acting on those motivations (e.g., antisocial personality traits, cognitions supportive of sexual offending), and situational factors are those that can provide an opportunity to offend (e.g., access to child victims). A meta-analysis found that men with CSEM-exclusive offending scored lower than men with contact-exclusive offending or men with mixed CSEM offending on measures of antisocial tendencies, hostility, criminal history, substance misuse, and unemployment (i.e., facilitative factors; Babchishin et al., 2015). Mixed offending men were more likely to be pedophilic – a motivational factor – than men with exclusive CSEM offences (Babchishin et al., 2015), who in turn were more likely to be pedophilic than men with exclusively contact sexual offences against children (Babchishin et al., 2015; see also Seto et al., 2006; Seto et al., 2017).

Recidivism studies of men with exclusively CSEM offences suggest that few will be adjudicated for contact sexual offences in the future, whereas mixed offending men are more likely to reoffend with contact sexual offences. For example, a UK study with an average follow-up of 13 years found that 3% of the 584 men with CSEM-exclusive offending had a new conviction for a contact sexual offence compared to 9% of the 106 mixed offending men (Elliott et al., 2019). A large cohort study of 4,658 men with CSEM offending found that, after 5 years of opportunity, 0.3% of men with CSEM-exclusive sexual offences committed a new contact sexual offence compared to 6% of men with mixed CSEM offending (Goller et al., 2010; Graf & Dittmann, 2011). To put this in perspective, the recidivism rates of men with CSEM-exclusive offending are similar to the sexual recidivism rates of men adjudicated for non-sexual offences, which is approximately 1 to 2% after 5 years (N = 543,024; Kahn et al., 2017). In contrast, the 5-year contact sexual recidivism rate of men with mixed CSEM offending (6% to 8%; Eke et al., 2019; Goller et al., 2010) is comparable to the rate of men with typical contact sexual offending; that is, men who have an average risk score on the Static-99R (score of 1 to 3; Static-99R [Helmus et al., 2012], the most commonly used actuarial risk assessment for men convicted of sexual offences; Kelley et al., 2020; Neal & Grisso, 2014) who have expected 5-year sexual recidivism rates from 4% to 8% (Hanson, Babchishin, et al., 2017).

Following the Risk, Need, and Responsivity (RNR) principles of correctional rehabilitation (Andrews & Bonta, 2010; Bonta & Andrews, 2016), individuals with a lower risk to reoffend should receive little to no correctional treatment (see also Hanson, Bourgon, et al., 2017 for suggested dosage thresholds). In line with principles of effective correctional rehabilitation, the majority of treatment programs designed for CSEM offending are similar to contact sexual offending programs in treatment targets but have lower treatment dosage (for review, see Paquette et al., 2020). Treatment evaluations for individuals with CSEM offending that examined reoffending, however, have found no difference in



offending between treatment and control groups (Beier et al., 2015; Wild et al., 2020; see also Table A.8 in Mews et al., 2017). One potential explanation is that the programs that target those with any CSEM offences overlook the possibility that individuals with CSEM offending are heterogeneous in risk of reoffending.

The order of the offences in mixed CSEM offending – whether a CSEM offence was committed before or after a contact sexual offence – and its association with risk factors and future reoffending has not yet been examined. Greater offence severity is generally unrelated to recidivism for individuals adjudicated for sexually motivated offences (Hanson & Bussière, 1998); however, severity predicts recidivism in other offending populations. For example, index offence severity predicted short-term recidivism (1 year) in a sample of adolescents adjudicated for criminal offences (e.g., Mulder et al., 2010), the severity of index offences predicts reoffending as well as the speed of reoffending in individuals adjudicated for interpersonal violence (Goldstein et al., 2016; Hilton & Eke, 2017; Hilton et al., 2004), and two large cohort studies of individuals incarcerated in federal institutions in Canada found that offence severity is predictive of readmission to Canadian federal custody (Offence Severity Record [OSR] of the Static Factor Assessment; Helmus & Forrester, 2017; Perley-Robertson et al., 2019).

Current Study

The current study examined the extent to which the temporal order of sexual offences by men who have committed CSEM offences is risk-relevant. Using the first two sexual crimes in the criminal record, we expected to find three patterns¹: *de-escalation pattern*, that is, men whose first sexual offence was a contact sexual offence followed by a CSEM offence; *Stable pattern*, that is men who committed a CSEM offence and whose next sexual offence was a CSEM offence; and, (3) *escalation pattern*, that is men who committed a CSEM offence and whose next sexual offence was a contact sexual offence. Cases where someone committed an offence producing CSEM with a child were categorized as a contact offence. Exploratory analyses were conducted to examine if CSEM groups differed on risk-relevant factors, as well as 5- and 7-year recidivism rates (start of follow-up: opportunity after the second sexual offence). Given the aim of the study is to examine CSEM offending, we did not examine the stable contact offending (contact to contact) pattern.



¹⁾ Terminology is an important consideration. The current study uses terminology common to the gateway hypothesis literature for ease of understanding. It is not meant to devalue the severity of CSEM offending.

Method

Participants

The current study presents a re-analysis of a sample of 387 men adjudicated for CSEM offences prior to 2011 (i.e., possessing, distributing, accessing, or making CSEM) in a large Canadian province previously reported in Eke et al. (2019), but with an extended follow-up period until 2018 (also used in Babchishin et al., 2022; average follow-up 20 years, range from 9 to 61 years following their first sexual crime). There was no preselection of cases; all available closed CSEM case files that met the original project criteria were included if there was sufficient information for the current study. We defined the index offences as the first two sexual offences in their criminal record, to determine order patterns (see Figure 1). On average, the individuals were 39 years (SD = 13; range = 18 to 76) old at release from their first CSEM offence.

Figure 1

Group Selection Flow-Chart (N = 387)



Note. CSEM = Child Sexual Exploitation Materials.

^aAll groups had to have at least one CSEM offence; those who did not have a CSEM offence in their first two sexual offences were excluded. Allegations were included in the groupings. Some individuals had both CSEM and contact offences at the same occasion; the most severe offence (contact) was used for the groupings.



There were 255 individuals whose first ever sexual offence was a CSEM offence and 97 individuals whose first ever sexual offence was a contact sexual crime. There were also 35 individuals whose first crime was another type of sexual crime (e.g., non-contact sexual crime, excluding CSEM); these cases were excluded from the current study. Of those remaining individuals whose first sexual crime was a CSEM or a contact sexual crime (n = 352), two-thirds (n = 229) did not have a second CSEM or contact sexual crime during the follow-up period. Of the 123 with two sexual offending occasions, 38 individuals whose first crime was contact sexual offence subsequently committed a sexual contact offence and were removed from the dataset because this group (contact to contact sexual offences) was not relevant to the purpose of the current study.

Of the 85 remaining men with two sexual offending occasions, the most common (47%, n = 40) pattern was a *stable pattern* (committed a CSEM offence and reoffended with a CSEM offence as their second reoffence), 41% (n = 35) fit a *de-escalation pattern* (committed a contact sexual offence and subsequently committed a CSEM offence as their next sexual offence); and 12% (n = 10) fit an *escalation* pattern (i.e., had a CSEM offence and subsequently reoffended with a contact sexual offence).

Procedure

The complete offending timeline and history of men with CSEM offending were available through to 2018. Offence history and offence details, demographic information, and psychological variables (available to a lesser extent) were coded from police file information in two previous studies (Eke et al., 2019; Seto & Eke, 2015). The criminal history spanned the criminal record, using police occurrence reports, the Canadian Police Information Centre (CPIC), and investigative file information from the collaborating police services in the original research. Charges and convictions were used to assign individuals into offending patterns, but we also used allegations (i.e., suspect or police investigation cases, but without resulting charge) to assign participants into groups when present (9% in the first offence [8/85], 8% in the second offence [7/85]). In cases where the sexual offence cluster included both contact and CSEM offending (e.g., producing CSEM offence with contact sexual victims), we characterized this offence as a contact offence (7/45 men with contact sexual offences; 16%). The original scoring of demographic variables and psychological variables were used, which had good interrater reliability in the original study (r = .94 to 1.00, kappa ranging from .70 to 1.00; Seto & Eke, 2015). The interrater reliability of the extended follow-up data was also good (ICC = .96 to 1.00 [Mdn = 1.00]; κ ranged from .50 to 1.00, *Mdn* = .93; Babchishin et al., 2022).

Measures

The archival database was administrative and contained demographic information, criminal history, and some psychological variables. We sorted these into two main domains:



items indicative of antisocial tendencies and items indicative of sexual interest in children. The archival dataset also included the Child Pornography Offender Risk Tool (CPORT, Seto & Eke, 2015): a 7-item structured tool to assess the likelihood of future sexual offending among men with CSEM offences. We rescored the CPORT to reflect the CPORT score at the time of the individual's second sexual offence – the offence used to create the individual's offending pattern. As with the method used to group individuals, information about allegations was available and we report CPORT scores with and without the inclusion of allegations of prior offending (allegations are not currently included in the formal scoring of the tool.²

Antisocial Tendencies

Age at police involvement (i.e., age at first police involvement, age at first CSEM offence), unemployed at the first CSEM offence, prior criminal history (i.e., total number of offences, and total number of any violent offences *prior* to their first sexual offence; these criminal history variables included allegations), substance misuse items (the Violence Risk Appraisal Guide [VRAG; Harris et al., 1993] items: severe drug use problem, severe alcohol use problem, and driving under the influence charge scored at first CSEM offence).

Sexual Interest in Children

Never married at the time of the first CSEM offence, number of sexual crimes in their index clusters (total number of CSEM offences including making CSEM offences, total number of contact sexual offences, total number of contact sexual victims) were extracted from the dataset. CPORT items indicative of sexual interest (CPORT Item 5 Pedophilia or hebephilia, CPORT Item 6 More boy than girl CSEM, and CPORT Item 7 More boy nude and other images) were recoded to reflect the second CSEM offence, when applicable.

Recidivism

Recidivism following their second sexual offence was primarily based on official charges or convictions but also included allegations (i.e., suspect or police investigation cases, but without resulting charge). We examined six types of recidivism: (1) CSEM offence, (2) contact sexual offence, (3) any sexual offence – any offence that was considered sexually motivated, (4) non-sexual violent offences, (5) violent offence – all crimes that involved direct confrontation with the victim (including contact sexual offences, but excluding non-contact sexual offences and sexually motivated breaches), and (6) any offence (sexual, violent, non-violent). To allow for comparisons with the published literature, we



²⁾ see https://www.researchgate.net/project/Child-Pornography-Offender-Risk-Tool-CPORT

computed 5-year recidivism rates. We also present 7-year follow-up results because we had sufficient data to do so.

Data Analysis

Men adjudicated for CSEM offences were classified according to their offending patterns. We then compared these groups on indicators of antisocial tendencies and sexual interest in children. We privileged effect sizes for interpretation of these exploratory analyses. Cohen's *d* was calculated using the formula from Cohen (1988) for continuous variables and estimated from dichotomous variables using the formula from Sánchez-Meca et al. (2003) and adding 0.5 to each cell to allow for *d* calculation with empty cells (Fleiss, 1994). Cohen's *d*s of .20, .50, and .80 are considered small, moderate, and large effects (Rice & Harris, 2005). Next, we compared the recidivism rates of the groups. The sample sizes for the recidivism analyses are small, as we required at least five and seven years following their latest sexual offence used for grouping.

Results

Group Differences

The de-escalation group had the youngest age at first police involvement and more prior offending than the escalation group (absolute *ds* ranging from .12 to .31) and stable group (absolute *ds* ranging from .27 to .32; see Table 1). The escalation group had more alcohol use, as well as charges for driving under the influence than the de-escalation group (absolute *ds* ranging from 0.39 to 0.58) and the stable group (absolute *ds* ranging from .85 to 1.20). The stable group scored lowest on substance use issues and had the fewest number of prior offences among the three groups. The stable group was also the least likely to be unemployed.

Indicators of sexual interest in children were based on criminal history, on characteristics of their CSEM collection (Items 6 and 7 of the CPORT), and on their self-reported or prior diagnosis of sexual interest in children (Item 5 of the CPORT). Unsurprisingly, both the escalation and de-escalation groups had more contact sexual offending victims than the stable group, and the stable group had more CSEM offences than the two mixed offending groups given our grouping definition. The stable group had more making CSEM offences than the de-escalation (absolute d = 0.32) and escalation groups (absolute d = 0.34). The Canadian legal definition for 'making' CSEM is broader than the legal definition of 'production' in the United States and included morphing images, reproducing images (e.g., saving images to CDs, printing text stories), writing text stories about sex with children, or creating websites for sharing CSEM. None of the making charges for the stable group included documentation of abuse.



Mixed De-Escalation:Contact \rightarrow CSEMVariableArisocial TendenciesAntisocial TendenciesAge at first police involvement27.5 (9.8), $n = 35$ Age at first CSEM offenceStart first CSEM offence	: Mixed Escalation: CSEM → Contact				
VariableAntisocial TendenciesAge at first police involvement $27.5 (9.8), n = 35$ Age at first CSEM offence $38.4 (13.2), n = 35$ Unemployed at the first CSEM offence $553 (14.3\%)$		Stable: CSEM → CSEM	Mixed Escalation vs Mixed De-Escalation ^a	Mixed Escalation vs Stable ^b	Mixed De-Escalation vs Stable ^c
Antisocial Tendencies $27.5 (9.8), n = 35$ Age at first police involvement $27.5 (9.8), n = 35$ Age at first CSEM offence $38.4 (13.2), n = 35$ Unemployed at the first CSEM offence $5/35 (14.3\%)$	$M\left(SD ight)$ or $n/N\left(\% ight)$			d [95% CI]	
Age at first police involvement $27.5 (9.8), n = 35$ Age at first CSEM offence $38.4 (13.2), n = 35$ Unemployed at the first CSEM offence $5/35 (14.3\%)$					
Age at first CSEM offence $38.4 (13.2)$, $n = 35$ Unemployed at the first CSEM offence $5/35 (14.3\%)$	32.2(15.5), n = 10	30.4 (11.4), n = 39	-0.42 [-1.12, 0.29]	-0.15 [-0.84, 0.55]	0.27 [-0.19, 0.73]
Unemployed at the first CSEM offence 5/35 (14.3%)	38.0(12.2), n = 9	33.1 (10.9), n = 40	0.03 [-0.70, 0.76]	-0.44 [-1.17 , 0.29]	-0.44 [$-0.90, 0.02$]
	1/10 (10.0%)	2/40 (5.0%)	-0.08 [-1.26, 1.10]	0.54 $[-0.76, 1.84]$	0.31 [-0.72, 1.33]
Frior number of non-sexual violent offences $0.46 (0.92)$, $n = 35$	0.20 (0.42), n = 10	0.23 (0.48), n = 40	-0.31 $[-1.01, 0.40]$	-0.05 [-0.75, 0.64]	0.32 [-0.13, 0.78]
Prior number of offences $2.20 (4.65)$, $n = 35$	1.70(2.45), n = 10	1.20(2.49), n = 40	-0.12 [-0.82, 0.59]	0.20 $[-0.49, 0.90]$	0.27 [-0.18, 0.73]
CPORT Total Score 3.26 (1.26), n = 34	р	3.19(1.31), n = 32	I	I	0.06 [-0.42, 0.54]
CPORT Total Score with the use of allegations $3.51 (1.15)$, $n = 35$	3.40(0.84), n = 10	3.05(1.36), n = 40	-1.10 $[-0.80, 0.60]$	0.27 [-0.42, 0.97]	0.37 [-0.09, 0.82]
Severe drug use problem (VRAG item) 0/15 (0.0%)	0/5 (0.0%)	0/23 (0.0%)	I	I	I
Severe alcohol use problem (VRAG item) 2/19 (10.5%)	1/6 (16.7%)	1/23 $(4.3%)$	0.39 [-0.97, 1.75]	0.85 [-0.63, 2.34]	0.46 [-0.82, 1.75]
Driving under the influence (DUJ) 5/35 (14.3%)	3/10 (30.0%)	2/40 (5.0%)	0.58 [-0.37, 1.52]	1.20 [0.10, 2.29]	0.62 [-0.33, 1.57]
Indicative of Interest in Children					
Total number of CSEM offences at index $1.17 (0.38)$, $n = 35$	1.10(0.32), n = 10	2.00(0.00), n = 40	-0.19 $[-0.90, 0.51]$	-6.58 [-8.04, -5.11]	-3.18 $[-3.86, -2.49]$
Making CSEM offences at index $0.37 (0.60)$, $n = 35$	0.20 (0.42), n = 10	0.58 (0.68), n = 40	-0.30 $[-1.01, 0.40]$	-0.34 [$-1.03, 0.36$]	-0.32 [-0.77, 0.14]
Never married at the first CSEM offence $16/33 (48.5\%)$	3/10 (30.0%)	23/38 (60.5%)	-0.43 $[-1.30, 0.44]$	-0.71 [-1.58, 0.15]	-0.29 [-0.85, 0.28]
Volunteered with kids 3/34 (8.8%)	0/10 (0.0%)	4/37~(10.8%)	-0.69 $[-2.51, 1.34]$	-0.63 [-2.45, 1.19]	-0.12 [-1.01, 0.78]
CPORT Item 5 (pedophilia or hebephilia) 15/32 (46.9%)	6/8 (75.0%)	31/34 (91.2%)	0.65 [-0.32, 1.63]	-0.75[-1.86, 0.36]	-1.41 [-2.19, -0.62]
CPORT Item 6 (more boy than girl CSEM) 3/33 (9.1%)	0/10 (0.0%)	10/37 (27.0%)	-0.53 [-2.38, 1.31]	-1.26 [-3.03, 0.51]	-0.73 [-1.53, 0.07]
CPORT Item 7 (more boy nude and other images) 4/31 (12.9%)	0/10 (0.0%)	10/37~(27.0%)	-0.74 [-2.57, 1.07]	-1.26 [-3.03, 0.51]	-0.51 [-1.26, 0.23]
Number of minor victims in contact sexual offence $0.77 (0.43)$, $n = 35$	0.60 (0.56), n = 10	0.00 (0.00), n = 40	-0.38 [-1.09, 0.32]	2.68 [1.81, 3.55]	2.65 [2.03, 3.27]

Sexual Offending: Theory, Research, and Prevention 2022, Vol. 17, Article e7229 https://doi.org/10.5964/sotrap.7229

Characteristics of CSEM Recidivists as Function of Order of Offending

Table 1

group. ^bMixed escalation group was the reference group; positive ds indicate that mixed escalation group had more risk-relevant problems than the stable group. Mixed de-escalation group was the reference group; positive ds indicate that mixed de-escalation group had more risk-relevant problems than the stable group.

'Only one participant scoring a 4 was in this cell; thus, ds for CPORT total score that included this group (i.e., mixed escalation) was not computed.

The stable group had more indicators of sexual interest in children based on their CSEM collections (more boy than girl images), reported sexual interest in children (or had a prior diagnosis for pedophilic disorder or hebephilic disorder as Paraphilic Disorder Not Otherwise Specified), and volunteered with children more than both mixed groups (absolute *ds* ranging from .12 to 1.41). The stable group, however, was more likely to be married (60.5%) than both the de-escalation and escalation groups (48.5% and 30%, respectively).

Recidivism

We examined 5-year and 7-year recidivism rates, following release from the latest sexual offence in the index offending cluster (see Table 2). The escalation group had the highest 5- and 7-year observed recidivism rates of contact sexual recidivism (14.3% for both follow-up times) compared to the de-escalation group (6.5% at 5-year, 12.9% at 7 years) and stable group (8.3% for both follow-up times). The escalation group also had the highest any crime recidivism (57.1% at both follow-up times) and non-sexual violent recidivism rates (28.6% at both follow-up times) but showed the lowest recidivism rates of CSEM offences (14.3%).

Table 2

	Any crime (sexual, violent, non-violent)		Non-sexual Violent		Violent (including contact sexual)		Any Sexual		Contact Sexual		CSEM	
Groups	5-year	7-year	5-year	7-year	5-year	7-year	5-year	7-year	5-year	7-year	5-year	7-year
De-Escalation:	9/31	11/31	2/31	3/31	3/31	6/31	9/31	11/31	2/31	4/31	7/31	7/31
(Contact \rightarrow CSEM)	(29.0%)	(35.5%)	(6.5%)	(9.7%)	(9.7%)	(19.4%)	(29.0%)	(35.5%)	(6.5%)	(12.9%)	(22.6%)	(22.6%)
Escalation:	4/7	4/7	2/7	2/7	3/7	3/7	2/7	2/7	1/7	1/7	1/7	1/7
(CSEM \rightarrow Contact)	(57.1%)	(57.1%)	(28.6%)	(28.6%)	(42.9%)	(42.9%)	(28.6%)	(28.6%)	(14.3%)	(14.3%)	(14.3%)	(14.3%)
Stable:	8/24	9/24	0/24	0/24	2/24	2/24	8/24	9/24	2/24	2/24	6/24	8/24
(CSEM \rightarrow CSEM)	(33.3%)	(37.5%)	(0.0%)	(0.0%)	(8.3%)	(8.3%)	(33.3%)	(37.5%)	(8.3%)	(8.3%)	(25.0%)	(33.3%)

Observed 5-Year and 7-Year Recidivism Rates of Three CSEM Recidivist Groups

Note. Four men with de-escalation (Contact \rightarrow CSEM) CSEM offending, 3 with Escalation (CSEM \rightarrow contact) CSEM offending, and 16 with stable (CSEM \rightarrow CSEM) offending did not have a 7-year follow-up time following their latest sexual offence. Such cases, thus, were excluded when calculating the recidivism rates. Recidivism events were primarily based on charges and convictions but also included police allegations.

The CSEM recidivism rates were the highest for the stable group (25% at 5-year, 33% at 7-year), followed by de-escalation group (23% for both 5-year and 7-year; Figure 2).



Figure 2

Observed 5- (Upper) and 7-Year (Lower) Recidivism Rates



Note. For the 5-year follow-up, recidivism event counts that were based on allegations and not official charges or convictions were: one violent offence allegation, two contact sexual offence allegations, and one CSEM offence allegation. For the 7-year follow-up, there were two violent allegations, three contact sexual allegations, and one CSEM allegation. There were no allegations for non-sexual violent offences in our sample.

Discussion

Our study adds to the literature that finds that CSEM offending is not a gateway to contact sexual offending for most men with CSEM offences. After an average 20-year follow-up from first sexual crime (follow-up range of 9 to 61 years), two-thirds of men



with CSEM offences were not detected for a reoffence with a contact sexual or CSEM offence. Of the 85 men with at least two sexual crimes during this time period, 5 in 10 fit the stable pattern, 4 in 10 fit the de-escalation pattern, and 1 in 10 fit the escalation pattern.

We conducted analyses to identify risk profiles and recidivism rates (5- and 7-years, following the opportunity after the second occasion for a sexual crime) of the three groups. Using a representative cohort of men with CSEM offending, we found that men who escalate with a new contact sexual offence after a CSEM offence tend to score higher on indicators of substance use than those who reoffend with a CSEM offence. We found that the de-escalation group had an earlier age of onset than the two other groups, had more prior criminal history, and fewer indicators of sexual interest. The stable group were the least likely to be unemployed and more likely to be married (despite scoring higher on sexual interest in children and having greater indicators of sexual interest in children than the other groups). Employment and marriage have been found to be protective against sexual offending (Laws & Ward, 2011). The current study suggests that these factors protect against contact sexual offending, but may not protect against CSEM offending, because 33% of the stable group reoffended with another CSEM offence after seven years.

Mixed sexual offending men have higher recidivism rates than men with CSEM-exclusive offending pattern (e.g., Elliott et al., 2019). The current study found that among mixed offending men, men who fit the escalation pattern are at a greater risk to reoffend with any crime, non-sexual violent offences, violent offences, and contact sexual offences than men who fit the de-escalation pattern, as well as those who fit the stable pattern. As a function of our methodology – all individuals had to have at least two sexual offences – observed 5- and 7-year recidivism rates were quite high for all three groups; indeed, the stable group had a higher 5-year contact sexual recidivism rate (8%) than typical samples of individuals with CSEM exclusive offending (typically 0-2%; see Babchishin et al., 2018, for review). In other words, having more than one sentencing occasion for a CSEM offence is a risk factor for future contact sexual offending, as well as further CSEM offences.

In sum, the temporal order of sexual offending is risk-relevant for CSEM offending. Men with CSEM offences who then committed a contact sexual offence (i.e., escalation pattern) were at a higher risk to reoffend than those with a stable pattern (CSEM to CSEM offending) and de-escalation pattern (contact to CSEM offending). Previous research has identified two distinct groups of individuals with CSEM offences: CSEM-exclusive (sexual offences are exclusively CSEM offences) and mixed CSEM offending (individuals with contact sexual offences and CSEM offences). The current study suggests that the number and ordering of sexual offences is risk-relevant for CSEM offending. Mixed CSEM offending men are heterogeneous in their risk profile and their risk to



reoffend and some of this heterogeneity can be explained by the number and order of sexual offences in their criminal history.

Practical Implications

Correctional programs that tailor intervention intensity to the risk profiles of participants are more effective at reducing reoffending (Andrews & Bonta, 2010; Bonta & Andrews, 2016). The current study found that men with CSEM offending, including mixed CSEM offending, vary on risk for recidivism; one size does not fit all individuals with CSEM offending. First, men with CSEM offending and an additional sexual crime (whether contact or CSEM) have recidivism rates similar to or higher than men with typical contact sexual offending, necessitating greater treatment intensity than men with a single CSEM offence. Second, among mixed offending individuals, those who fit the escalation pattern are at higher risk to reoffend than those who fit the de-escalation pattern. Treatment programs that allow for varying intensity of treatment among individuals with CSEM offending may provide better recidivism reduction than programs that treat all individuals with CSEM offending similarly. Such programs have yet to find a statistically significant reduction in recidivism (Beier et al., 2015; Wild et al., 2020; also see Table A.8 in Mews et al., 2017). Risk tools can be used to help sort individuals with CSEM offending in terms of their estimated risk to reoffend (for review, see Brown, 2022).

Among those with at least two sexual offending occasions, most fit the stable or de-escalation pattern, with 1 in 10 fitting the escalation pattern. The escalation group had more indicators of substance use and had higher rates of recidivism for all but CSEM reoffending. As such, men with CSEM offences who also have substance use problems may be particularly at risk to commit a new contact sexual offence. However, additional prospective research studies that include substance use measures are needed to replicate this finding.

Nonsexual recidivism rates, whether violent or nonviolent, were notable for all groups. As such, general recidivism tools (in addition to those designed to predict sexual recidivism) could be useful for individuals with CSEM offending. For a proportion of individuals with CSEM offending, treatment should include interventions that target general criminality. Men with mixed CSEM offending may particularly benefit from general interventions. In contrast, the stable group may benefit most from interventions that target and help manage their atypical sexual interests. All groups likely would benefit from potentially facilitating factors associated with CSEM use (e.g., coping, use of leisure time; Seto, 2013). Given the heterogeneity in CSEM offending observed in the current study, general criminality (e.g., antisocial personality traits, antisocial friend) and sexual criminality (e.g., atypical sexual interests, sexual preoccupation, sexualized coping; Mann et al., 2010) assessments would be valuable in directing interventions (e.g., supervision, programming).



Limitations and Future Directions

A large sample size and lengthy follow-up are required to investigate the offending pattern of men with CSEM offending. Despite a follow-up period from the first recorded sexual offence ranging from 9 to 61 years (M = 20 years), two-third of our sample of men with CSEM offending were not detected for a new contact sexual or CSEM offence. The analyses were based on the one-third that did commit a reported second sexual offence during the follow-up period (n = 85), and most of these were classified in the stable and de-escalation groups. As a result, our analyses (especially recidivism analyses) were based on a small sample size. The small sample size precluded us from examining the predictive accuracy of the CPORT for these subgroups. Future research aiming to examine the predictive accuracy of risk tools for mixed CSEM subgroups would likely need to combine datasets because most men with CSEM offending are expected to exclusively have one detected CSEM offence or fit the stable group (two CSEM offences).

We also supplemented official charges and conviction with allegations to sort individuals into offending groups. This meant that 8 individuals (7 contact to CSEM and 1 CSEM to CSEM) were sorted based solely on allegations, without charges or convictions. Reasons for allegations to be closed without charges were lack of supporting or reliable evidence, the suspect's identity could not be confirmed, or the case was closed as "unsolved". Many of these allegations were older, from the 1980s or early 1990s, and involves cases where the full police reports were no longer available. Although we used allegations to inform group allocation for only a minority of the sample (8 of 85), it is possible that the use of allegations could have reduced group differences to the extent that the allegations were incorrect and resulted in misclassification. We also included individuals with mixed offending in the contact group. Namely, if someone had both CSEM and contact sexual offences in the same sentencing occassion, we classified this individual in the contact sexual offence group. Future research, with larger sample sizes, may be able to examine these dual offending occassions (contact and CSEM in the same index) as a separate group. It is possible that men that follow a mixed stable pattern (i.e., contact and CSEM offence followed by another contact and CSEM offence) are particularly risky.

The current study drew criminal information from official criminal records, which is not a complete representation of all offences committed by a person. In addition, not all charges or convictions are included or submitted to the national system used in the current study, including some lesser or "summary" offences, offences diverted from the criminal justice system, and offences committed outside of Canada. Some charges or convictions can be purged, including offences committed as a youth or offences as an adult that were pardoned. Juvenile offending is therefore expected to be underrepresented in our sample, in part because the original sampling strategy necessitated that each individual had an adult conviction for CSEM. Also, youth offending data were known if noted in the file information, rather than it being based on an examination



of juvenile records, which are sealed or purged if an individual remains offence-free as an adult for a number of years, as specified in the Canadian *Youth Criminal Justice Act.* For those individuals in our sample who were younger at the time of their index CSEM offence, there is perhaps a greater likelihood that juvenile data would be noted in an investigative file. Overall then, age at first police involvement may have been missing some juvenile offences and may be better described as age of first police involvement in adulthood. Studies have found that earlier age of onset is related to more extensive criminal trajectory (e.g., Le Blanc & Loeber, 1998; Moffitt & Caspi, 2003; Piquero & Moffitt, 2008). Common criticisms of age of onset studies include mainly methodological concerns such as the threshold between early and late onset being sample specific and inconsistent across studies and offence type being ignored despite studies finding that the first offence type is meaningful (Tzoumakis et al., 2013). In the current study, the de-escalation group were about 5 years younger than the escalation group and three years younger than the stable group. This finding is limited as our study did not have a complete coverage of criminal offences committed during adolescents for our sample.

One potential explanation for our patterns of findings is that the three groups may not be meaningfully different and instead are being examined at different stages in their criminal trajectory. Specifically, the current study used a cohort of men varying in age. Although we could look at criminal history, we only looked forward until 2018. It is possible that some men with a single sexual offence may have gone on to commit a new sexual offence should we have more complete information. Indeed, the stable group were five years younger than the escalation and de-escalation group at their first CSEM offences. In addition, although some escalation men committed new contact sexual offences, others committed new CSEM offences or committed no offence after 5to 7-years follow-up. In other words, classification may change with time. Given that samples of individuals with CSEM offending are reaching longer follow-up times (e.g., > 20 years), more nuanced pictures of reoffending and stability of offending can finally be elucidated. Future research on CSEM offending should code the ordering of CSEM offending so that they can contribute to this line of research. The current study had a long follow-up from first offence (average of 20 years) and therefore could go beyond their initial sexual recidivism events. The sample size, however, was small and precludes strong conclusions, especially for recidivism rates.

The current study used a dataset based on available police reports and, as a result, the majority of our variables were criminal history and demographic indicators. Psychologically meaningful measures would contribute greatly to our understanding of the etiology and course of CSEM offending. In particular, it would be interesting to see how cognitions specific to CSEM offending and situational factors (e.g., alcohol use during the offence) may predict reoffending among individuals with CSEM offending (for review, see Steel et al., 2020). Mixed offending may be a promising indicator of antisociality as the current study found that both mixed groups (de-escalation and escalation) were high-



er on our indicators of antisociality than the stable group. Our antisociality indicators (substance use, criminal history), however, were quite limited. Future research would benefit from additional indicators of antisocial tendencies and other risk-relevant propensities. Such studies would be beneficial to direct treatment and management efforts of individuals adjudicated with CSEM offences. We also did not collect information on self-reported offending, and systematic reviews suggest a larger proportion of CSEM have contact sexual offences if based on self-report rather than official records (Seto et al., 2011). We were able to include allegations (i.e., suspect cases and allegation that resulted in police investigations), as well as arrests, charges, or convictions for crimes in our definitions of recidivism.

Conclusion

The current study took advantage of the longer follow-up now available for CSEM studies to explore if the ordering of contact sexual and CSEM offending is risk-relevant among men with CSEM offending. We found that the ordering of CSEM offences is risk-relevant; mixed offending men are not all the same. Using the first two sexual crimes in their criminal trajectory to define groups, we found that escalation, de-escalation, and stable groups differ on risk-relevant indicators and 5- and 7-year recidivism rates. The escalation group had greater substance use issues, the de-escalation group had a younger age of onset and more prior offending, and the stable group had more indicators of sexual interest in children. We also found that a significant proportion (8%) of men who fit the stable pattern will go on to commit a contact sexual offence after 5- and 7-years. Management and treatment of individuals with CSEM offending should therefore be sensitive to their sexual offending patterns.

Funding: The authors have no funding to report.

Acknowledgments: We would like to thank members of the Ontario Provincial Police's Threat and Behaviour Analysis Team and Grace Culp for their comments on an earlier draft of the manuscript. The views expressed in this paper are that of the authors, and are not necessarily shared by their organization.

Competing Interests: The authors have declared that no competing interests exist.

Twitter Accounts: @KBabchishin, @seungclee, @MCSeto

Data Availability: Data are not publicly available due to existing data agreements.



References

- Adams, W., & Flynn, A. (2017). Federal prosecution of commercial sexual exploitation of children cases, 2004-2013 (NCJ 25076). US Department of Justice, Bureau of Justice Statistics. https://www.bjs.gov/content/pub/pdf/fpcsecc0413.pdf
- Andrews, D. A., & Bonta, J. (2010). The psychology of criminal conduct (5th ed.). Routledge.
- Babchishin, K. M., Eke, A. W., Lee, S. C., Lewis, N., & Seto, M. C. (2022). Applying offending trajectory analyses to men adjudicated for child sexual exploitation material offenses. *Criminal Justice and Behavior*, 49(8), 1095–1114. https://doi.org/10.1177/00938548211040849
- Babchishin, K. M., Hanson, R. K., & Hermann, C. A. (2011). The characteristics of online sex offenders: A meta-analysis. *Sexual Abuse*, 23(1), 92–123. https://doi.org/10.1177/1079063210370708
- Babchishin, K. M., Hanson, R. K., & VanZuylen, H. (2015). Online child pornography offenders are different: A meta-analysis of the characteristics of online and offline sex offenders against children. Archives of Sexual Behavior, 44(1), 45–66. https://doi.org/10.1007/s10508-014-0270-x
- Babchishin, K. M., Merdian, H. L., Bartels, R. M., & Perkins, D. (2018). Child sexual exploitation materials offenders: A review. *European Psychologist*, 23(2), 130–143. https://doi.org/10.1027/1016-9040/a000326
- Beier, K. M., Grundmann, D., Kuhle, L. F., Scherner, G., Konrad, A., & Amelung, T. (2015). The German Dunkelfeld project: A pilot study to prevent child sexual abuse and the use of child abusive images. *Journal of Sexual Medicine*, 12(2), 529–542. https://doi.org/10.1111/jsm.12785
- Bonta, J., & Andrews, D. A. (2016). The psychology of criminal conduct. Routledge.
- Brown, S. J. (2022). Assessing the risk of users of child sexual exploitation material committing further offences: A scoping review. *Journal of Sexual Aggression*. Advance online publication. https://doi.org/10.1080/13552600.2022.2104394
- Cohen, J. (1988). *Statistical power analysis for the behavioral sciences* (2nd ed.). Lawrence Erlbaum Associates.
- Department of Justice Canada. (2019). *Just Facts: Sexual violations against children and child pornography*. https://www.justice.gc.ca/eng/rp-pr/jr/jf-pf/2019/docs/mar02.pdf
- Eke, A. W., Helmus, L. M., & Seto, M. C. (2019). A validation study of the Child Pornography Offender Risk Assessment Tool (CPORT). Sexual Abuse, 31(4), 456–476. https://doi.org/10.1177/1079063218762434
- Elliott, I. A., Mandeville-Norden, R., Rakestrow-Dickens, J., & Beech, A. R. (2019). Reoffending rates in a U.K. community sample of individuals with convictions for indecent images of children. *Law and Human Behavior*, 43(4), 369–382. https://doi.org/10.1037/lhb0000328
- Europol, European Union Agency for Law Enforcement Cooperation. (2020). *Exploiting isolation: Offenders and victims of online child sexual abuse during the COVID-19 pandemic.* https://www.europol.europa.eu/sites/default/files/documents/europol_covid_reportcse_jun2020v.3_0.pdf
- Fleiss, J. L. (1994). Measures of effect size for categorical data. In H. Cooper & L. V. Hedges (Eds.), Handbook of research synthesis (pp. 245–260). Russell Sage Foundation.



- Goldstein, D. A., Cantos, A. L., Brenner, L. H., Verborg, R. J., & Kosson, D. S. (2016). Perpetrator type moderates the relationship between severity of intimate partner violence and recidivism. *Criminal Justice and Behavior*, 43(7), 879–898. https://doi.org/10.1177/0093854815616841
- Goller, A., Graf, M., Frei, A., & Dittmann, V. (2010). Recidivism of internet sex offenders An epidemiologic study on more than 4600 offenders in Switzerland. *European Psychiatry*, 25(Suppl. 1), Article 1533. https://doi.org/10.1016/S0924-9338(10)71516-9
- Graf, M., & Dittmann, V. (2011). Forensic-psychiatric treatment for internet sex offenders. In D. P. Boer, R. Eher, L. A. Craig, M. H. Miner, & F. Pfäfflin (Eds.), *International perspectives on the* assessment and treatment of sexual offenders: Theory, practice, and research (pp. 479–488). Wiley. https://doi.org/10.1002/9781119900420.ch25
- Hamilton, M. (2011). The efficacy of severe child pornography sentencing: Empirical validity or political rhetoric? *Stanford Law & Policy Review*, 22(2), 545–586.
 https://law.stanford.edu/publications/efficacy-severe-child-pornography-sentencing-empirical-validity-political-rhetoric
- Hanson, R. K., Babchishin, K. M., Helmus, L. M., Thornton, D., & Phenix, A. (2017). Communicating the results of criterion referenced prediction measures: Risk categories for the Static-99R and Static-2002R sexual offender risk assessment tools. *Psychological Assessment*, 29(5), 582–597. https://doi.org/10.1037/pas0000371
- Hanson, R. K., Bourgon, G., McGrath, R., Kroner, D., D'Amora, D. A., Thomas, S. S., & Tavarez, L. P. (2017). A five-level risk and needs system: Maximizing assessment results in corrections through the development of a common language. The Council of State Governments Justice Center. https://csgjusticecenter.org/wp-content/uploads/2020/01/A-Five-Level-Risk-and-Needs-System_Report.pdf
- Hanson, R. K., & Bussière, M. T. (1998). Predicting relapse: A meta-analysis of sexual offender recidivism studies. *Journal of Consulting and Clinical Psychology*, 66(2), 348–362. https://doi.org/10.1037/0022-006X.66.2.348
- Harris, G. T., Rice, M. E., & Quinsey, V. L. (1993). Violent recidivism of mentally disordered offenders: The development of a statistical prediction instrument. *Criminal Justice and Behavior*, 20(4), 315–335. https://doi.org/10.1177/0093854893020004001
- Helmus, L. M., & Forrester, T. (2017). Validating the predictive accuracy of the Static Factors Assessment (SFA) risk scale for federally sentenced offenders in Canada. *Canadian Journal of Criminology and Criminal Justice*, 59(1), 1–25. https://doi.org/10.3138/cjccj.2015.E04
- Helmus, L., Thornton, D., Hanson, R. K., & Babchishin, K. M. (2012). Improving the predictive accuracy of Static-99 and Static-2002 with older sex offenders: Revised age weights. *Sexual Abuse*, 24(1), 64–101. https://doi.org/10.1177/1079063211409951
- Henshaw, M., Ogloff, J. R., & Clough, J. A. (2018). Demographic, mental health, and offending characteristics of online child exploitation material offenders: A comparison with contact-only and dual sexual offenders. *Behavioral Sciences & the Law*, 36(2), 198–215. https://doi.org/10.1002/bsl.2337



- Hilton, N. Z., & Eke, A. W. (2017). Assessing risk of intimate partner violence. In J. C. Campbell & J. T. Messing (Eds.), Assessing dangerousness: Domestic violence offenders and child abusers (pp. 139–178). Springer. https://doi.org/10.1891/9780826133274.0006
- Hilton, N. Z., Harris, G. T., Rice, M. E., Lang, C., Cormier, C. A., & Lines, K. J. (2004). A brief actuarial assessment for the prediction of wife assault recidivism: The Ontario domestic assault risk assessment. *Psychological Assessment*, 16(3), 267–275. https://doi.org/10.1037/1040-3590.16.3.267
- Kahn, R. E., Ambroziak, G., Hanson, R. K., & Thornton, D. (2017). Release from the sex offender label. Archives of Sexual Behavior, 46(4), 861–864. https://doi.org/10.1007/s10508-017-0972-y
- Kandel, D. B., & Logan, J. A. (1984). Patterns of drug use from adolescence to young adulthood: I. Periods of risk for initiation, continued use, and discontinuation. *American Journal of Public Health*, 74(7), 660–666. https://doi.org/10.2105/AJPH.74.7.660
- Kelley, S. M., Ambroziak, G., Thornton, D., & Barahal, R. M. (2020). How do professionals assess sexual recidivism risk? An updated survey of practices. *Sexual Abuse*, 32(1), 3–29. https://doi.org/10.1177/1079063218800474
- Laws, D. R., & Ward, T. (2011). Desistance from sex offending: Alternatives to throwing away the keys. Guilford Press.
- Le Blanc, M., & Loeber, R. (1998). Developmental criminology updated. *Crime and Justice, 23*, 115–198. https://doi.org/10.1086/449270
- Loeber, R., & Hay, D. (1997). Key issues in the development of aggression and violence from childhood to early adulthood. *Annual Review of Psychology*, 48(1), 371–410. https://doi.org/10.1146/annurev.psych.48.1.371
- Mann, R. E., Hanson, R. K., & Thornton, D. (2010). Assessing risk for sexual recidivism: Some proposals on the nature of psychologically meaningful risk factors. *Sexual Abuse*, 22(2), 191– 217. https://doi.org/10.1177/1079063210366039
- McManus, M. A., & Almond, L. (2014). Trends of indecent images of children and child sexual offenses between 2005/2006 and 2012/2013 within the United Kingdom. *Journal of Sexual Aggression*, 20(2), 142–155. https://doi.org/10.1080/13552600.2014.893031
- Mews, A., Di Bella, L., & Purver, M. (2017). *Impact evaluation of the prison-based core sex offender treatment programme* (Ministry of Justice Analytical Series). London, United Kingdom: Ministry of Justice.

 $https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/623876/sotp-report-web-.pdf$

- Moffitt, T. E., & Caspi, A. (2003). Preventing the inter-generational continuity of antisocial behaviour: Implications of partner violence. In D. P. Farrington & J. W. Coid (Eds.), *Early prevention of adult antisocial behaviour* (pp. 109–129). Cambridge University Press.
- Mulder, E., Brand, E., Bullens, R., & Van Marle, H. (2010). A classification of risk factors in serious juvenile offenders and the relation between patterns of risk factors and recidivism. *Criminal Behaviour and Mental Health*, 20(1), 23–38. https://doi.org/10.1002/cbm.754



- Neal, T. M. S., & Grisso, T. (2014). Assessment practices and expert judgment methods in forensic psychology and psychiatry: An international snapshot. *Criminal Justice and Behavior*, 41(12), 1406–1421. https://doi.org/10.1177/0093854814548449
- Paquette, S., Fortin, F., & Perkins, D. (2020). Online sexual offenders: Typologies, assessment, treatment, and prevention. In J. Proulx, F. Cortoni, L. A. Craig, & E. J. Letourneau (Eds.), The Wiley handbook of what works with sexual offenders: Contemporary perspectives in theory, assessment, treatment, and prevention (pp. 311–326). https://doi.org/10.1002/9781119439325.ch18
- Perley-Robertson, B., Helmus, L. M., & Forth, A. (2019). Predictive accuracy of static risk factors for Canadian Indigenous offenders compared to non-Indigenous offenders: Implications for risk assessment scales. *Psychology, Crime & Law, 25*(3), 248–278. https://doi.org/10.1080/1068316X.2018.1519827
- Piquero, A. R., & Moffitt, T. E. (2008). Explaining the facts of crime: How the developmental taxonomy replies to Farrington's invitation. In D. P. Farrington (Ed.), *Integrated developmental* and life-course theories of offending (pp. 51–72). Transaction.
- Prescott, C. A. (1976). Genetic influences on stages of substance use and addiction. In H. D. Chilcoat (Ed.), Understanding the pathway from use to addiction (pp. 27–28). National Institute on Drug Abuse (NIDA).
- Rice, M. E., & Harris, G. T. (2005). Comparing effect sizes in follow-up studies: ROC Area, Cohen's d, and r. Law and Human Behavior, 29(5), 615–620. https://doi.org/10.1007/s10979-005-6832-7
- Sánchez-Meca, J., Marín-Martínez, F., & Chacón-Moscoso, S. (2003). Effect-size indices for dichotomized outcomes in meta-analysis. *Psychological Methods*, 8(4), 448–467. https://doi.org/10.1037/1082-989X.8.4.448
- Seto, M. C. (2013). Internet sex offenders. American Psychological Association.
- Seto, M. C., Cantor, J. M., & Blanchard, R. (2006). Child pornography offenses are a valid diagnostic indicator of pedophilia. *Journal of Abnormal Psychology*, 115(3), 610–615. https://doi.org/10.1037/0021-843X.115.3.610
- Seto, M. C., & Eke, A. W. (2015). Predicting recidivism among adult male child pornography offenders: Development of the Child Pornography Offender Risk Tool (CPORT). *Law and Human Behavior*, *39*(4), 416–429. https://doi.org/10.1037/lhb0000128
- Seto, M. C., Hanson, R. K., & Babchishin, K. M. (2011). Contact sexual offending by men with online sexual offenses. Sexual Abuse, 23(1), 124–145. https://doi.org/10.1177/1079063210369013
- Steel, C. M., Newman, E., O'Rourke, S., & Quayle, E. (2020). A systematic review of cognitive distortions in online child sexual exploitation material offenders. *Aggression and Violent Behavior*, 51(1), Article 101375. https://doi.org/10.1016/j.avb.2020.101375
- Seto, M. C., Stephens, S., Lalumière, M. L., & Cantor, J. M. (2017). The revised screening scale for pedophilic interests (SSPI–2): Development and criterion-related validation. *Sexual Abuse*, 29(7), 619–635. https://doi.org/10.1177/1079063215612444
- Steiker, C. S. (2013). Lessons from two failures: Sentencing for cocaine and child pornography under the federal sentencing guidelines in the United States. *Law and Contemporary Problems*, 76(1), 27–52. https://scholarship.law.duke.edu/lcp/vol76/iss1/3



- Tzoumakis, S., Lussier, P., Le Blanc, M., & Davies, G. (2013). Onset, offending trajectories, and crime specialization in violence. *Youth Violence and Juvenile Justice*, 11(2), 143–164. https://doi.org/10.1177/1541204012458440
- Wild, T. S. N., Müller, I., Fromberger, P., Jordan, K., Klein, L., & Müller, J. L. (2020). Prevention of sexual child abuse: Preliminary results from an outpatient therapy program. *Frontiers in Psychiatry*, 11, Article 88. https://doi.org/10.3389/fpsyt.2020.00088
- Wright, D. (2020). COVID-19: Expectations and effects on children online. UK Safer Internet Centre. https://swgfl.org.uk/assets/documents/covid-19-expectations-and-effects-on-childrenonline.pdf
- Yamaguchi, K., & Kandel, D. B. (1984). Patterns of drug use from adolescence to young adulthood: II. Sequences of progression. *American Journal of Public Health*, 74(7), 668–672. https://doi.org/10.2105/AJPH.74.7.668



Sexual Offending: Theory, Research, and Prevention (SOTRAP) is the official journal of the International Association for the Treatment of Sexual Offenders (IATSO).



PsychOpen GOLD is a publishing service by Leibniz Institute for Psychology (ZPID), Germany.

